Author's response to reviews

Title: Improving smoking cessation support for pregnant and postpartum women: an exploratory study

Authors:

Tracey Borland (tracey.borland@camh.ca)
Alexey Babayan (alexey.babayan@utoronto.ca)
Saeeda Irfan (saeeda.irfan@cancercare.on.ca)
Robert Schwartz (Robert.Schwartz@utoronto.ca)

Version: 2  Date: 16 January 2013

Author's response to reviews:

Dear Editors and staff of BMC Public Health:

In response to your request for making changes to the formatting of our manuscript, the following sentence has been added in the Methods section (at the end of the first paragraph):

"Written consent was obtained from all participants of the study prior to interviews".

Thanks for considering our submission.

Sincerely,
Alexey Babayan (on behalf of the research team)

Alexey Babayan, PhD
Ontario Tobacco Research Unit
Dalla Lana School of Public Health
University of Toronto
530 – 155 College St
Toronto, ON M5T 3M7
416-978-7096 (P)
416-946-0340 (F)
alexey.babayan@utoronto.ca